Author's response to reviews

Title: A survey of medical students to assess their exposure to and knowledge of renal transplantation

Authors:

Anusha G Edwards (dr_ag_edwards@yahoo.com)
Andrew R Weale (andy@weale.org.uk)
Justin D Morgan (justin.morgan@north-brisol.swest.nhs.uk)

Version: 3 Date: 22 December 2004

Author's response to reviews:

Dear Emma

Thank you for your email of 21st December regarding the acceptance, in principle, of the revised manuscript entitled "A survey of medical students to assess their exposure to and knowledge of renal transplantation".

As requested we have made the formatting changes that you specified in your email. Regarding the reviewers comments, we have addressed one of the discretionary revisions regarding the actual numbers of transplants performed at Bristol and Plymouth. These figures have been included in the first paragraph of the results section.

We are happy to submit this manuscript as a final version but please do let us know if there are any further changes that need to be made.

May we take this opportunity to wish you and your team a Happy Christmas.

Yours sincerely

Anusha Edwards
Andrew Weale
Justin Morgan